\documentclass[12pt]{article} % BB Nov 26 2011.
\usepackage{epsfig}
\textheight 9.0in \textwidth  6.0in \topmargin -0.4in \headsep 0.4in  
\oddsidemargin  0.25in  \evensidemargin 0.25in  

\def\sqr#1#2{{\vcenter{\hrule height.#2pt
   \hbox{\vrule width.#2pt height#1pt \kern#1pt
   \vrule width.#2pt} \hrule height.#2pt}}}
\def\square{{\mathchoice\sqr64\sqr64\sqr{4.2}3\sqr{3.0}3}}
\def\undertext#1{$\underline{\smash{\hbox{#1}}}$}
\pagestyle{empty}

\begin{document}

\noindent Solution 34: 
\begin{enumerate}
\item Marion and Thornton p.409.
\item Marion and Thornton p.421.
\item Yes. Marion and Thornton p.425.
\end{enumerate}
 
\end{document}

